Dear Editor

We are sending the manuscript Treatment of uterine leiomyomata in adolescence for analysis of your Journal.

Despite is a systematic review of literature regarding these cases, the occurrence of these tumors in adolescence are not common and management is scarcely discussed in the literature, So, we believe this will make a good contribution.

Kind regards

Luiz Gustavo Brito (on behalf of the authors)